Author's response to reviews

Title: The burden of disease profile of residents of Nairobi's slums: Results from a Demographic Surveillance System

Authors:

Catherine Kyobutungi (ckyobutungi@aphrc.org)
Abdhalah K Ziraba (aziraba@aphrc.org)
Alex Ezeh (aezeh@aphrc.org)
Yazoume Ye (yyazoume@aphrc.org)

Version: 5 Date: 1 March 2008

Author's response to reviews:

Formmatting changes requested:
1) Table and figure placement text deleted
2) Titles on both figures deleted
3) Figures cropped
In addition:
4) Typing and content errors corrected and minor changes made to text